Author's response to reviews

Title: A study of best positive predictors for sustained virologic response to interferon alpha plus ribavirin therapy in naive chronic hepatitis C patients

Authors:

Muhammad Idrees (idreeskhan96@yahoo.com)
Sheikh Riazuddin (riaz@ihr.comsat.edu.pk)

Version: 5 Date: 12 January 2009

Author's response to reviews:

The Biomed Central Editorial Team

Object: MS: MS: 2037671906198646- “A study of best positive predictors for sustained virologic response to interferon alpha plus ribavirin therapy in naive chronic hepatitis C patients”. Dr. Muhammad Idrees and Dr. Sheikh Riazuddin.

Thank you for accepting our manuscript for publication in your esteemed journal.

We have made changes in the format of the above manuscript according to your suggestions. It may now be straightforward for online production process.

Best regards

Idrees